Author's response to reviews

Title: The effects of a web-based decision aid on the intention to diagnostic self-testing for cholesterol and diabetes: A randomized controlled trial

Authors:

Gaby Ronda (gaby.ronda@maastrichtuniversity.nl)
Janaica EJ Grispen (janaica@hotmail.com)
Martine HP Ickenroth (martineickenroth@hotmail.com)
Geert-Jan Dinant (geertjan.dinant@maastrichtuniversity.nl)
Nanne K De Vries (nanne.devries@maastrichtuniversity.nl)
Trudy Van der Weijden (trudy.vanderweijden@maastrichtuniversity.nl)

Version: 4
Date: 29 August 2014

Author's response to reviews: see over
August 2014

To the editors of BMC Public Health

Reference number: BMC Public Health, MS: 1048484725118261

Dear Editors,

Please find enclosed our manuscript ‘The effects of a web-based decision aid on the intention to diagnostic self-testing for cholesterol and diabetes: A randomized controlled trial’ (BMC Public Health, MS: 1048484725118261).

There was one remaining editorial request.

1.) Please specifically mention all authors in the Authors' Contribution section.

We now mentioned all authors with their initials in the Authors' Contribution section (page 20).

On behalf of all the authors,

Sincerely,

Gaby Ronda